Author's response to reviews

Title: Traditional Serrated Adenoma of the Sigmoid Colon with Osseous metaplasia. A Case Report.

Authors:

Nelson Montalvo (nmontalvof@gmail.com)
José Beltrán (josribens@hotmail.com)
Ligia Redrobán (ligiredroban@yahoo.es)

Version: 5 Date: 6 March 2012

Author's response to reviews:

1. Please move the tables in your paper after the References section.
   Ok, we do it.

2. Please format your title to follow the structure: A presenting in B with C: a case report.
   Previous title: Osseous Metaplasia in a Traditional Serrated Adenoma of the sigmoid colon. A case report.
   Modified title:
   Traditional Serrated Adenoma of the sigmoid colon with osseous metaplasia. A case report.